Author's response to reviews

Title: Individual socio-demographic factors and perceptions of the environment as determinants of inequalities in adolescent physical and psychological health: the Olympic Regeneration in East London (ORiEL) study

Authors:

Neil R Smith (n.r.smith@qmul.ac.uk)
Daniel J Lewis (daniel.lewis@lshtm.ac.uk)
Amanda Fahy (amanda.fahy@qmul.ac.uk)
Sandra Eldridge (s.eldridge@qmul.ac.uk)
Stephanie J.C. Taylor (s.j.c.taylor@qmul.ac.uk)
Derek G Moore (d.g.moore@uel.ac.uk)
Charlotte Clark (c.clark@qmul.ac.uk)
Stephen A Stansfeld (s.a.stansfeld@qmul.ac.uk)
Steven Cummins (steven.cummins@lshtm.ac.uk)

Version: 2
Date: 30 July 2014

Author's response to reviews:

As requested, the following revisions were made prior to editorial assessment:

1. Tables 1 to 8 were moved the end of document to immediately follow the references section.

2. The authors' contribution section was reformatted to more explicitly detail which author was responsible for which duty.

3. A paragraph has been inserted within the ethics section to clarify that parental consent was passive in accordance with the wishes of the Ethics Board, and that the participants each provided active assent.